\documentclass[11pt,twoside]{article}
\usepackage{amsmath, amsthm, amscd, amsfonts, amssymb, graphicx, color}
\usepackage[bookmarksnumbered, colorlinks]{hyperref} \usepackage{float}
\usepackage{lipsum}
\usepackage{afterpage}
\usepackage[labelfont=bf]{caption}
\usepackage[nottoc,notlof,notlot]{tocbibind} 
%\renewcommand\bibname{References}
\def\bibname{\Large \bf  References}
\usepackage{lipsum}
\usepackage{fancyhdr}
\pagestyle{fancy}
\fancyhf{}
\renewcommand{\headrulewidth}{0pt}
\fancyhead[LE,RO]{\thepage}
\thispagestyle{empty}
%\afterpage{\lhead{new value}}

\fancyhead[CE]{G. Farid and V. N. Mishra} 
\fancyhead[CO]{Caputo fractional derivative inequalities via $(h-m)$-convexity}



%\topmargin=-1.6cm
\textheight 17.5cm%
\textwidth  12cm %
\topmargin   8mm  %
\oddsidemargin   20mm   %
\evensidemargin   20mm   %
\footskip=24pt     %

\newtheorem{theorem}{Theorem}[section]
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{corollary}[theorem]{Corollary}
\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{example}[theorem]{Example}
\newtheorem{xca}[theorem]{Exercise}
%\theoremstyle{remark}
\newtheorem{remark}[theorem]{Remark}
\renewenvironment{proof}{{\bfseries \noindent Proof.}}{~~~~$\square$}
\makeatletter
\def\th@newremark{\th@remark\thm@headfont{\bfseries}}
\makeatletter



  
  
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% If you want to insert other packages. Insert them here
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%\long\def\symbolfootnote[#1]#2{\begingroup%
%\def\thefootnote{\fnsymbol{footnote}}\footnote[#1]{#2}\endgroup}



 \def \thesection{\arabic{section}}
 

\begin{document}
\begin{thebibliography}{99} % Enter references in alphabetical order and according to the following format.

%
\bibitem{PS}  
G. A. Anastassiou, {\it Advances on fractional inequalities}, NY, Springer, 2011.
\bibitem{A} D. Assante, C. Cesarano, C. Fornaro and L. Vazquez, {\it Higher Order and
Fractional Diffusive Equations}, JESTR  8 (2015), 202-204, ISSN: 1791-2377.
 
\bibitem{PS1} 
G. A. Anastassiou, {\it Fractional differentiation inequalities}, NY, Springer, Heidelberg, 2009. 
\bibitem{B} C. Cesarano, {\it Generalized special functions in the description of
fractional diffusive equations}, CAIM, 10 (2019), 31-40, ISSN: 2038-0909.

\bibitem{gf} G. Farid, \emph{Some Riemann-Liouville fractional integral inequalities for convex functions}, J. Anal., (2018), doi.org/10.1007/s41478-0079-4.

\bibitem{sgf} G. Farid, W. Nazeer, S. M. Saleem, S. Mehmood and S. M. Kang, \emph{Bounds of Riemann-Liouville fractional integrals in general form via convex functions and their applications}, Math. 2018(6) (2018), 248. 

\bibitem {AA}
A. A. Kilbas, H. M. Srivastava, J. J. Trujillo, {\it Theory and Applications of fractional differential equations}, North-Holland Mathematics Studies, 204, Elsevier, New York-London, 2006.

\bibitem {MLK}
M. Lazarevi\'{c}, {\it Advanced Topics on Applications of Fractional Calculus on Control Problems}, System stability and modeling, WSEAS Press, 2014.

\bibitem {p09} 
C. P. Li, W. H. Deng, {\it Remarks on fractional derivatives}, Appl. Math. Comput. 187(2) (2007), 777-784.


\bibitem{ref6} 
M.E. \"{O}zdemir, A.O. Akdemri, E. Set, \it{ On $(h-m)$-convexity and Hadamard-type inequalities}, Transylv. J. Math. Mech. 8(1) (2016), 51-58.

\bibitem{hconvex}  S. Varosanec, {\it On $h$-convexity}, J. Math. Anal. Appl. 326(1) (2007), 303--311.
\end{thebibliography}

%\baselineskip 9mm
%\setcounter{page}{}
\thispagestyle{plain}
{\noindent Journal of Mathematical Extension \\
Vol. XX, No. XX, (2014), pp-pp (Will be inserted by layout editor)}\\
ISSN: 1735-8299\\
URL: http://www.ijmex.com\\
‎\vspace*{9mm}
‎
\begin{center}

{\Large \bf 
Caputo fractional derivative inequalities via $(h-m)$-convexity\\}
 


\let\thefootnote\relax\footnote{\scriptsize Received: XXXX; Accepted: XXXX (Will be inserted by editor)}

{\bf Ghulam Farid}\vspace*{-2mm}\\
\vspace{2mm} {\small  COMSATS University Islamabad, Attock Campus, Pakistan} \vspace{2mm}

{\bf  Vishnu Narayan Mishra$^*$\let\thefootnote\relax\footnote{$^*$Corresponding Author}}\vspace*{-2mm}\\
\vspace{2mm} {\small Department of Mathematics, Indira Gandhi National Tribal University, Lalpur, Amarkantak, Anuppur, Madhya Pradesh 484 887, India} \vspace{2mm}

\end{center}

\vspace{4mm}


{\footnotesize
\begin{quotation}
{\noindent \bf Abstract.} The aim of this study is to establish some new Caputo fractional integral inequalities. By applying definition of $(h-m)$-convexity and some straightforward inequalities an upper bound of the sum of left and right sided Caputo fractional derivatives has been established. Furthermore, a modulus inequality and a Hadamard type inequality have been analyzed. These results provide various fractional inequalities for all particular functions deducible from $(h-m)$-convexity, see Remark \ref{rem1}.
\end{quotation}
\begin{quotation}
\noindent{\bf AMS Subject Classification:} 26A51; 26A33; 26D15

\noindent{\bf Keywords and Phrases:} Convex function, $(h-m)$-convex function, Caputo fractional derivatives
\end{quotation}}

\section{Introduction}
\label{intro} % It is advised to give each section and subsection a unique label.
Nobody can deny the importance of fractional calculus in the field of engineering, fluid mechanics, mathematical analysis etc. Many mathematicians have been introduced many articles using fractional calculus (see \cite{A, B, gf, sgf} and references therein).\\
The study on the fractional calculus continued with the contributions from Fourier, Abel, Lacroix, Leibniz, Grunwald and Letnikov for detail (see, \cite{AA,MLK}). In the realm of the fractional differential equations, Caputo fractional derivative and Riemann-Liouville ones are mostly used. They generalize the ordinary integral and differential operators. However, the fractional derivatives have fewer properties than the
corresponding classical ones. On the other hand, besides the smooth requirement,
Caputo derivative does not coincide with the classical derivative \cite{p09}. For detail of fractional derivatives readers are suggested \cite{PS,PS1,AA}. 
\begin{definition}\cite{AA}
	Let $\alpha>0$ and $\alpha \notin\{1, 2, 3, . . . \}$, $n = [\alpha]+1$,  $f \in AC^n[a,b]$, the space of functions having $nth$ derivatives absolutely continuous. Then the left-sided and right-sided Caputo fractional derivatives of order $\alpha$ are defined as follows: 
	\begin{align}
	&({^C}D^{\alpha}_{a+}f)(x)=\frac{1}{\Gamma(n-\alpha)}\int^{x}_{a}\frac{f^{(n)} {(t)}}{(x-t)^{{\alpha}-n+1}}dt ,        x>a
	\end{align}
	and
	\begin{align}
	&({^C}D^{\alpha}_{b-}f)(x)=\frac{(-1)^n}{\Gamma(n-\alpha)}\int^{b}_{x}\frac{f^{(n)} {(t)}}{(t-x)^{{\alpha}-n+1}}dt ,         x<b.
	\end{align}
	If $\alpha = n \in \{1, 2, 3, . . . \}$ and usual derivative $f^{(n)}(x)$ of order $n$ exists, then Caputo fractional derivative $({^C}D^{n}_{a+}f)(x)$ coincides with $f^{(n)}(x)$ whereas $({^C}D^{n}_{b-}f)(x)$ coincides with $f^{(n)}(x)$ with exactness to a constant multiplier $(-1)^{n}$. 
	In particular we have
	\begin{align}
	&({^C}D^{0}_{a+}f)(x)=({^C}D^{0}_{b-}f)(x)= f(x)
	\end{align}
	where $n=1$ and $\alpha=0$.
\end{definition}
\indent The aim of this paper is to find fractional inequalities for the Caputo fractional derivatives via $(h-m)$-convex  functions. The $(h-m)$-convexity is the generalization of convexity on right half of the real line including zero (see, \cite{ref6, hconvex} and references therein). Moreover, these fractional inequalities appear as a compact formulation which contain various induced results for all functions deducible from $(h-m)$-convex functions, see Remark \ref{rem1}.
\begin{definition}			
	Let $J\subseteq \mathbb{R}$ be an interval containing $(0,1)$ and let $h : J\rightarrow \mathbb{R}$ be a non-negative function. We say that $f :[0,b]\rightarrow\mathbb{R}$ is a $(h-m)$-convex function, if $f$ is non-negative and for all $x,y\in[0,b], m\in[0,1]$ and $\alpha\in(0,1)$, one has
	\begin{equation}\label{010*}
	f(\alpha x+m(1-\alpha)y)\leq h(\alpha)f(x)+mh(1-\alpha)f(y).
	\end{equation}
	If reverse of the inequality holds, then $f$ is called $(h-m)$-concave function. 
\end{definition}
For suitable choice of $h$ and $m$, class of $(h-m)$-convex functions is reduced to the different known classes of functions defined on $[0,b]$.
\begin{remark}\label{rem1}
	(i) By setting $m=1$ in \eqref{010*}, it reduces to the definition of $h$-convex function.\\
	(ii) By setting $h(\alpha)=\alpha$ in \eqref{010*}, it reduces to the definition of $m$-convex function.\\
	(iii) By setting $h(\alpha)=\alpha$ and $m=1$ in \eqref{010*}, it reduces to the definition of convex function.\\
	(iv) By setting $h(\alpha)=1$ and $m=1$ in \eqref{010*}, it reduces to the definition of $p$-function.\\
	(v) By setting $h(\alpha)=\alpha^s$ and $m=1$ in \eqref{010*}, it reduces to the definition of $s$-convex function of second sense.\\
	(vi) By setting $h(\alpha)=\frac{1}{\alpha}$ and $m=1$ in \eqref{010*}, it reduces to the definition of Godunova-Levin function.\\
	(vii) By setting $h(\alpha)=\frac{1}{\alpha^s}$ and $m=1$ in \eqref{010*}, it reduces to the definition of $s$-Godunova-Levin function of second kind.
\end{remark}
\section{Main Results}
Firstly, an upper bound of the sum of left and right Caputo fractional derivatives (CFD) has been proved by applying inequalities in the result of the definition of $(h-m)$-convex function. The established upper bound of (CFD) via $(h-m)$-convex function contains upper bounds in particular for $h$-convex, $m$-convex and convex functions, also for $s$-convex function of second sense, Godunova-Levin function and $p$-function. Further a modulus inequality is established for (CFD) by using $(h-m)$-convexity of $f^{(n+1)}$. At the end an inequality of Hadamard type is obtained. Moreover, presented results have been further studied in particular points in the domain.
\begin{theorem}\label{thm1}
	Let $f:[0,\infty)\rightarrow \mathbb{R}$ be a function such that $f\in AC^n[a,b]$, $0\leq a<b$. If $f^{(n)}$ is $(h-m)$-convex, then for $\alpha,\beta>1$ the following inequality for the Caputo fractional derivatives holds:
	\begin{align}\label{2.1}
	& \left(^{C}D_{a^{+}}^{\alpha-1}f\right)(x)+\left(^{C}D_{b^{-}}^{\beta-1}f\right)(x)\\&\leq \left(\frac{(x-a)^{n-\alpha+1}f^{(n)}(a)}{\Gamma(n-\alpha+1)}+\frac{(b-x)^{n-\beta+1}f^{(n)}(b)}{\Gamma(n-\beta+1)}\right.\nonumber\\&\left.+mf^{(n)}\left(\frac{x}{m}\right)\left(\frac{(b-x)^{n-\beta+1}}{\Gamma(n-\beta+1)}+\frac{(x-a)^{n-\alpha+1}}{\Gamma(n-\alpha+1)}\right)\right)\int_{0}^{1}h(z)dz.\nonumber
	\end{align}
\end{theorem}
\begin{proof}
	Using definition of $(h-m)$-convex function for $f^{(n)}$, we have
	\begin{equation}\label{1}
	f^{(n)}(t)\leq h\left(\frac{x-t}{x-a}\right)f^{(n)}(a)+mh\left(\frac{t-a}{x-a}\right)f^{(n)}\left(\frac{x}{m}\right),
	\end{equation}
	where we use the identity
	\begin{equation*}
	t=\frac{x-t}{x-a}a+m\frac{t-a}{x-a}\frac{x}{m}.
	\end{equation*}
	Also for $\alpha>0$ and $t\in[a,x]$, we have
	\begin{equation}\label{2}
	(x-t)^{n-\alpha}\leq(x-a)^{n-\alpha}.
	\end{equation}
	Multiplying (\ref{1}) and (\ref{2}) and integrating over $[a,x]$, we get
	\begin{align}\label{2.7}
	&\int_{a}^{x}(x-t)^{n-\alpha}f^{(n)}(t)dt\leq f^{(n)}(a)(x-a)^{n-\alpha}\int_{a}^{x}h\left(\frac{x-t}{x-a}\right)dt\\&+mf^{(n)}\left(\frac{x}{m}\right)(x-a)^{n-\alpha}\int_{a}^{x}h\left(\frac{t-a}{x-a}\right)dt.\nonumber
	\end{align}
	Now by using definition of Caputo fractional derivative on left hand side of (\ref{2.7}) and by change of variables on its right hand side, we get
	\begin{equation}\label{3}
	\left(^{C}D_{a^{+}}^{\alpha-1}f\right)(x)\leq\frac{(x-a)^{n-\alpha+1}}{	\Gamma(n-\alpha+1)}\left(f^{(n)}(a)+mf^{(n)}\left(\frac{x}{m}\right)\right)\int_{0}^{1}h(z)dz.
	\end{equation}
	Similarly using $(t-x)^{n-\beta}\leq(x-b)^{n-\beta}$,\, $t\in[x,b]$ and definition of $(h-m)$-convexity of $f^{(n)}$ for the identity
	\begin{equation*}
	t=\frac{t-x}{b-x}b+m\frac{b-t}{b-x}\frac{x}{m}
	\end{equation*}	
	one can have
	\begin{equation}\label{4}
	\left(^{C}D_{b^{-}}^{\beta-1}f\right)(x)\leq\frac{(b-x)^{n-\beta+1}}{\Gamma(n-\beta+1)}\left(f^{(n)}(b)+mf^{(n)}\left(\frac{x}{m}\right)\right)\int_{0}^{1}h(z)dz.
	\end{equation}
	Adding (\ref{3}) and (\ref{4}), we get the inequality in (\ref{2.1}).
\end{proof}
\begin{corollary}
	By taking $\alpha=\beta$ in (\ref{2.1}), we get the following inequality for Caputo fractional derivatives:
	\begin{align*}
	&\left(^{C}D_{a^{+}}^{\alpha-1}f\right)(x)+\left(^{C}D_{b^{-}}^{\alpha-1}f\right)(x)\\&\leq\frac{1}{\Gamma(n-\alpha+1)} \Big((x-a)^{n-\alpha+1}f^{(n)}(a)+(b-x)^{n-\alpha+1}f^{(n)}(b)\nonumber\\&+mf^{(n)}\left(\frac{x}{m}\right)\left((b-x)^{n-\alpha+1}+(x-a)^{n-\alpha+1}\right)\Big)\int_{0}^{1}h(z)dz.\nonumber
	\end{align*}
\end{corollary}	
\begin{remark}\label{rem2}
	(i) If we put $m=1$ in \eqref{2.1}, then bound of Caputo fractional derivative for $h$-convex function is established.\\
	(ii) If we put $h(\alpha)=\alpha$ in \eqref{2.1}, then bound of Caputo fractional derivative for $m$-convex function is established.\\
	(iii) If we put $h(\alpha)=\alpha$ and $m=1$ in \eqref{2.1}, then bound of Caputo fractional derivative for convex function is established..\\
	(iv) If we put $h(\alpha)=1$ and $m=1$ in \eqref{2.1}, then bound of Caputo fractional derivative for $p$-function is established..\\
	(v) If we put $h(\alpha)=\alpha^s$ and $m=1$ in \eqref{2.1}, then bound of Caputo fractional derivative for $s$-convex function of second sense is established.\\
	(vi) If we put $h(\alpha)=\frac{1}{\alpha}$ and $m=1$ in \eqref{2.1}, then bound of Caputo fractional derivative for Godunova-Levin function is established.\\
	(vii) If we put $h(\alpha)=\frac{1}{\alpha^s}$ and $m=1$ in \eqref{2.1}, then bound of Caputo fractional derivative for $s$-Godunova-Levin function of second kind is established.
\end{remark}
\begin{theorem}\label{thm2}
	Let $f:[0,\infty)\rightarrow \mathbb{R}$ be a function such that $f\in AC^{n+1}[a,b]$, $0\leq a<b$. If $|f^{(n+1)}|$ is $(h-m)$-convex, then for $\alpha,\beta>0$ the following inequality for the Caputo fractional derivatives holds:
	\begin{align}\label{0}
	&\left|(^{C}D^{\alpha}_{a^{+}}f)(x) + (^{C}D^{\beta}_{b^{-}}f)(x)-\left(\frac{(x-a)^{n-\alpha}f^{(n)}(a)}{\Gamma(n-\alpha + 1)} +\frac{(b-x)^{n-\beta}f^{(n)}(b)}{\Gamma(n-\beta + 1)}\right)\right|\\&\leq \left(\frac{(x-a)^{n-\alpha+1}|f^{(n+1)}(a)|}{\Gamma(n-\alpha + 1)}+\frac{(b-x)^{n-\beta+1}|f^{(n+1)}(b)|}{\Gamma(n-\beta + 1)}\nonumber\right.\\&\left. +m\left|f^{(n+1)}\left(\frac{x}{m}\right)\right|\left(\frac{(x-a)^{n-\alpha+1}}{\Gamma(n-\alpha + 1)}+\frac{(b-x)^{n-\beta+1}}{\Gamma(n-\beta + 1)}\right)\right)\int^{1}_{0} h(z)dz.\nonumber
	\end{align} 
\end{theorem}
\begin{proof}
	Since $|f^{(n+1)}|$ is $(h-m)$-convex, therefore for $t \in [a,x]$, we have
	\begin{equation*}
	|f^{(n+1)}(t)|\leq h\left(\frac{x-t}{x-a}\right)|f^{(n+1)}(a)|+mh\left(\frac{t-a}{x-a}\right)\left|f^{(n+1)}\left(\frac{x}{m}\right)\right|,
	\end{equation*}
	from which we can write
	\begin{align}\label{2.11}
	&-\left(h\left(\frac{x-t}{x-a}\right)|f^{(n+1)}(a)|+mh\left(\frac{t-a}{x-a}\right)\left|f^{(n+1)}\left(\frac{x}{m}\right)\right|\right)\\&\leq f^{(n+1)}(t) \leq h\left(\frac{x-t}{x-a}\right)|f^{(n+1)}(a)|+mh\left(\frac{t-a}{x-a}\right)\left|f^{(n+1)}\left(\frac{x}{m}\right)\right|.\nonumber
	\end{align}
	We consider the second inequality of (\ref{2.11}), that is
	\begin{equation}\label{s}
	f^{(n+1)}(t) \leq h\left(\frac{x-t}{x-a}\right)|f^{(n+1)}(a)|+mh\left(\frac{t-a}{x-a}\right)\left|f^{(n+1)}\left(\frac{x}{m}\right)\right|.
	\end{equation}
	Now for $\alpha>0$, we have the following inequality
	\begin{equation}\label{m}
	(x-t)^{n-\alpha} \leq (x-a)^{n-\alpha},\,\,\,\ t\in[a,x].
	\end{equation}
	Multiplying the last two inequalities and integrating with respect to $t$ over $[a,x]$, we have 
	\begin{align}\label{2.14}
	&\int_{a}^{x}(x-t)^{n-\alpha}f^{(n+1)}(t)dt\\&\leq(x-a)^{n-\alpha}\left[|f^{(n+1)}(a)|\int_{a}^{x}h\left(\frac{x-t}{x-a}\right)dt+m\left|f^{(n+1)}\left(\frac{x}{m}\right)\right|\right.\nonumber\\&\left.\int_{a}^{x}h\left(\frac{t-a}{x-a}\right)dt\right].\nonumber
	\end{align}
	The left hand side of (\ref{2.14}) is calculated as follows
	\begin{align*}
	\int_{a}^{x}(x-t)^{n-\alpha}f^{(n+1)}(t)dt=-f^{(n)}(a)(x-a)^{n-\alpha}+\Gamma(n-\alpha+1)(^{C}D^{\alpha}_{a^{+}}f)(x),
	\end{align*}
	while using change of variables in the right hand side of (\ref{2.14}), the resulting inequality takes the form
	as follows
	\begin{align}\label{x}
	&(^{C}D^{\alpha}_{a^{+}}f)(x)-\frac{f^{(n)}(a)(x-a)^{n-\alpha}}{\Gamma(n-\alpha+1)}\\&\leq\frac{(x-a)^{n-\alpha+1}}{\Gamma(n-\alpha+1)}\left(|f^{(n+1)}(a)|+m\left|f^{(n+1)}\left(\frac{x}{m}\right)\right|\right)\int^{1}_{0} h(z)dz.\nonumber
	\end{align}
	If we consider from (\ref{2.11}), the left hand side inequality and proceeding as we did for the right side inequality, we get
	\begin{align}\label{c}
	&\frac{f^{(n)}(a)(x-a)^{n-\alpha}}{\Gamma(n-\alpha+1)}-(^{C}D^{\alpha}_{a^{+}}f)(x)\\&\leq\frac{(x-a)^{n-\alpha+1}}{\Gamma(n-\alpha+1)}\left(|f^{(n+1)}(a)|+m\left|f^{(n+1)}\left(\frac{x}{m}\right)\right|\right)\int^{1}_{0} h(z)dz.\nonumber
	\end{align} 
	From (\ref{x}) and (\ref{c}), we get
	\begin{align}\label{0001}
	&\left|(^{C}D^{\alpha}_{a^{+}}f)(x)-\frac{f^{(n)}(a)(x-a)^{n-\alpha}}{\Gamma(n-\alpha+1)}\right|\\&\leq\frac{(x-a)^{n-\alpha+1}}{\Gamma(n-\alpha+1)}\left(|f^{(n+1)}(a)|+m\left|f^{(n+1)}\left(\frac{x}{m}\right)\right|\right)\int^{1}_{0} h(z)dz.\nonumber
	\end{align}
	On the other hand for $t\in[x,b]$ using $(h-m)$-convexity of $|f^{(n+1)}|$, we have
	\begin{equation}\label{001}
	|f^{(n+1)}(t)|\leq h\left(\frac{t-x}{b-x}\right)|f^{(n+1)}(b)|+mh\left(\frac{b-t}{b-x}\right)\left|f^{(n+1)}\left(\frac{x}{m}\right)\right|.
	\end{equation}
	Also for $t\in[x,b]$ and $\beta>0$, we have
	\begin{equation}\label{002}
	(t-x)^{n-\beta}\leq(b-x)^{n-\beta}.
	\end{equation}
	By adopting the same treatment as we have done for (\ref{2.11}) and (\ref{m}), one can obtain from (\ref{001}) and (\ref{002}), the following inequality
	\begin{align}\label{0002}
	&\left|(^{C}D^{\beta}_{b^{-}}f)^{(n)}(x)-\frac{f^{(n)}(b)(b-x)^{n-\beta}}{\Gamma(n-\beta+1)}\right|\\&\leq\frac{(b-x)^{n-\beta+1}}{\Gamma(n-\beta+1)}\left(|f^{(n+1)}(b)|+m\left|f^{(n+1)}\left(\frac{x}{m}\right)\right|\right)\int^{1}_{0} h(z)dz.\nonumber
	\end{align}
	By combining the inequalities (\ref{0001}) and (\ref{0002}) via triangular inequality, we get the required inequality.
\end{proof}
\begin{corollary}
	By taking $\alpha=\beta$ in (\ref{0}), then we get the following inequality for Caputo fractional derivatives: 
	\begin{align*}
	&\left|(^{C}D^{\alpha}_{a^{+}}f)(x) + (^{C}D^{\alpha}_{b^{-}})f(x)\right.\\&\left.-\frac{1}{\Gamma(n-\alpha + 1)}\left((x-a)^{n-\alpha}f^{(n)}(a) \!+\!(b-x)^{n-\alpha}f^{(n)}(b)\right)\right|\\&\leq \frac{1}{\Gamma(n-\alpha + 1)}\Big((x-a)^{n-\alpha+1}|f^{(n+1)}(a)|+(b-x)^{\alpha+1}|f^{(n+1)}(b)|\nonumber\\& +m\left|f^{(n+1)}\left(\frac{x}{m}\right)\right|\left((x-a)^{n-\alpha+1}+(b-x)^{n-\alpha+1}\right)\Big)\int^{1}_{0} h(z)dz.\nonumber
	\end{align*}
\end{corollary}
\begin{remark}
	Axioms (i)-(vii) of Remark \ref{rem2} for Theorem \ref{thm1} are valid for Theorem \ref{thm2}.
\end{remark}
%\begin{corollary}
%	By setting $\alpha=\beta=1$  and $x=\frac{a+b}{2}$ in (\ref{0}) we get the following inequality
%	\begin{align}\label{010}
%	&\left|\frac{1}{b-a}\int_{a}^{b}f(t)dt-\frac{f(a)+f(b)}{2}\right|\\&\leq\frac{b-a}{4}\left(|f'(a)|+|f'(b)|+2mf'\left(\frac{a+b}{2}\right)\right)\int^{1}_{0} h(z)dz.\nonumber
%	\end{align}
%\end{corollary}
%\begin{corollary}
%	By setting $\alpha=\beta=1$  and $x=\frac{a+b}{2}$ and $m=0$ in (\ref{0}) we get the following inequality
%	\begin{align}\label{010}
%	&\left|\frac{1}{b-a}\int_{a}^{b}f(t)dt-\frac{f(a)+f(b)}{2}\right|\leq\frac{b-a}{4}\left(|f'(a)|+|f'(b)|\right)\int^{1}_{0} h(z)dz.
%	\end{align}
%\end{corollary}
% \begin{remark}
% 	If we put $h(t)=t$, then from (\ref{010}) we get \cite[Theorem 2.2]{da}. It is interesting to see that by selecting appropriate function $h$ this inequality can be refined as we desire.
% \end{remark}
First of all we prove the following lemma.
\begin{lemma}\label{lemm}
	Let $f:[0,\infty)\rightarrow \mathbb{R}$ be a function such that $f\in AC^n[a,b]$, $0\leq a<b$. If $f^{(n)}$ is $(h-m)$-concave and $f^{(n)}\left(\frac{a+b-x}{m}\right)=f^{(n)}(x)$, then the following inequality holds:
	\begin{equation}\label{0110}
	f^{(n)}\left(\frac{a+b}{2}\right)\geq (m+1)h\left(\frac{1}{2}\right)f^{(n)}(x),\,x\in[a,b].
	\end{equation}
\end{lemma}
\begin{proof}
	We have
	\begin{equation}
	\frac{a+b}{2}=\frac{1}{2}\left(\frac{x-a}{b-a}b+\frac{b-x}{b-a}a\right)+\frac{1}{2}\left(\frac{x-a}{b-a}a+\frac{b-x}{b-a}b\right).
	\end{equation}
	Since $f^{(n)}$ is $(h-m)$-concave, therefore we have
	\begin{align}\label{D}
	&f^{(n)}\left(\frac{a+b}{2}\right)\\&\nonumber\geq h\left(\frac{1}{2}\right)\left[f^{(n)}\left(\frac{x-a}{b-a}b+\frac{b-x}{b-a}a\right)+mf^{(n)}\left(\frac{x-a}{m(b-a)}a+\frac{b-x}{m(b-a)}b\right)\right]\\&=h\left(\frac{1}{2}\right)\left(f^{(n)}(x)+mf^{(n)}\left(\frac{a+b-x}{m}\right)\right).\nonumber
	\end{align}
	Now by using the condition $f^{(n)}\left(\frac{a+b-x}{m}\right)=f^{(n)}(x)$, inequality in (\ref{0110}) can be obtained.
\end{proof}
\begin{theorem}\label{thm3}
	Let $f:[0,\infty)\rightarrow\mathbb{R}$ be a function such that $f\in AC^n[a,b]$, $0\leq a<b$. If $f^{(n)}$ is $(h-m)$-concave, then for $\alpha,\beta>0$ the following inequality for the Caputo fractional derivatives holds:
	\begin{align}\label{2.23}
	\left(mf^{(n)}\left(\frac{b}{m}\right)+f^{(n)}(a)\right)&\leq \frac{\Gamma(n-\beta)(^{C}D^{\beta}_{a^+}f)(b)}{(b-a)^{n-\beta}}+\frac{\Gamma(n-\alpha)(^{C}D^{\alpha}_{b^-}f)(a)}{(b-a)^{n-\alpha}}\\&\leq\frac{1}{(m+1)h\left(\frac{1}{2}\right)}\left(\frac{1}{n-\beta}+\frac{1}{n-\alpha}\right)f^{(n)}\left(\frac{a+b}{2}\right).\nonumber
	\end{align}
\end{theorem}
\begin{proof}
	For $\alpha,\beta>0$, $n-\alpha-1\leq0$ and  
	\begin{equation}\label{a}
	(b-a)^{n-\alpha-1}\leq(x-a)^{n-\alpha-1}.
	\end{equation}
	Since $f^{(n)}$ is $(h-m)$ concave, therefore we have
	\begin{equation}\label{b}
	mh\left(\frac{x-a}{b-a}\right)f^{(n)}\left(\frac{b}{m}\right)+h\left(\frac{b-x}{b-a}\right)f^{(n)}(a)\leq f^{(n)}(x).
	\end{equation}
	Multiplying (\ref{a}) with (\ref{b}) and then integrating over $[a,b]$, we get
	\begin{align*}
	&(b-a)^{n-\alpha-1}\left[mf^{(n)}\left(\frac{b}{m}\right)\int_{a}^{b}h\left(\frac{x-a}{b-a}\right)dx+f^{(n)}(a)\int_{a}^{b}h\left(\frac{b-x}{b-a}\right)dx\right]\\&\leq\int_{a}^{b}(x-a)^{n-\alpha-1}f^{(n)}(x)dx
	\end{align*}
	which implies
	\begin{equation}\label{d}
	(-1)^n\Gamma(n-\alpha)(^{C}D^{\alpha}_{b^-}f)(a)\geq(b-a)^{n-\alpha}\left(mf^{(n)}\left(\frac{b}{m}\right)+f^{(n)}(a)\right)\int_{0}^{1}h(z)dz.
	\end{equation}
	Also
	\begin{equation}\label{e}
	(b-a)^{n-\beta-1}\leq(b-x)^{n-\beta-1}.
	\end{equation}
	Multiplying (\ref{e}) with (\ref{b}), and then integrating over $[a,b]$, we get
	\begin{align*}
	&(b-a)^{n-\beta-1}\left[mf^{(n)}\left(\frac{b}{m}\right)\int_{a}^{b}h\left(\frac{x-a}{b-a}\right)dx+f^{(n)}(a)\int_{a}^{b}h\left(\frac{b-x}{b-a}\right)dx\right]\\&\leq\int_{a}^{b}(b-x)^{n-\beta-1}f^{(n)}dx
	\end{align*}
	which implies
	\begin{equation}\label{f}
	(-1)^n\Gamma(n-\beta)(^{C}D^{\beta}_{a^+}f)(a)\geq(b-a)^{n-\beta}\left(mf^{(n)}\left(\frac{b}{m}\right)+f^{(n)}(a)\right)\int_{0}^{1}h(z)dz.
	\end{equation}
	From (\ref{d}) and (\ref{f}), we get
	\begin{align}\label{g}
&\frac{\Gamma(n-\beta)(^{C}D^{\beta}_{a^+}f)(b)}{(b-a)^{n-\beta}}+\frac{\Gamma(n-\alpha)(^{C}D^{\alpha}_{b^-}f)(a)}{(b-a)^{n-\alpha}}\\&\nonumber\geq\left(mf^{(n)}\left(\frac{b}{m}\right)+f^{(n)}(a)\right)\int_{0}^{1}h(z)dz.
	\end{align}
	Now from Lemma 1, we have
	\begin{equation}\label{h}
	(m+1)h\left(\frac{1}{2}\right)f^{(n)}(x)\leq f^{(n)}\left(\frac{a+b}{2}\right).
	\end{equation}
	Multiplying (\ref{h}) by $(x-a)^{n-\alpha-1}$ and integrating over $[a,b]$, we have
	\begin{equation*}
	\int_{a}^{b}(x-a)^{n-\alpha-1}f^{(n)}(x)dx\leq\frac{f^{(n)}\left(\frac{a+b}{2}\right)}{(m+1)h\left(\frac{1}{2}\right)}\int_{a}^{b}(x-a)^{n-\alpha-1}dx,
	\end{equation*}
	\begin{equation}\label{i}
	\Gamma(n-\alpha)(^{C}D^{\alpha}_{b^-}f)(a)\leq\frac{(b-a)^{n-\alpha}}{(n-\alpha)(m+1)h\left(\frac{1}{2}\right)}f^{(n)}\left(\frac{a+b}{2}\right).
	\end{equation}
	Multiplying (\ref{h}) by $(b-x)^{n-\beta-1}$ and integrating over $[a,b]$, we have
	\begin{equation*}
	\int_{a}^{b}(b-x)^{n-\beta-1}f^{(n)}(x)dx\leq\frac{f^{(n)}\left(\frac{a+b}{2}\right)}{(m+1)h\left(\frac{1}{2}\right)}\int_{a}^{b}(b-x)^{n-\beta-1}dx,
	\end{equation*}
	\begin{equation}\label{j}
	\Gamma(n-\beta)(^{C}D^{\beta}_{a^+}f)(x)\leq\frac{(b-a)^{n-\beta}}{(n-\beta)(m+1)h\left(\frac{1}{2}\right)}f^{(n)}\left(\frac{a+b}{2}\right).
	\end{equation}
	Adding (\ref{i}) and (\ref{j}), we obtain
	\begin{align}\label{k}
	&\frac{\Gamma(n-\beta)(^{C}D^{\beta}_{a^+}f)(b)}{(b-a)^{n-\beta}}+\frac{\Gamma(n-\alpha)(^{C}D^{\alpha}_{b^-}f)(a)}{(b-a)^{n-\alpha}}\\&\leq\frac{1}{(m+1)h\left(\frac{1}{2}\right)}\left(\frac{1}{n-\beta}+\frac{1}{n-\alpha}\right)f^{(n)}\left(\frac{a+b}{2}\right).\nonumber
	\end{align}
	From (\ref{g}) and (\ref{k}), we get required inequality.
\end{proof}	
\begin{corollary}
	By taking $\alpha=\beta$ in (\ref{2.23}), then we get the following inequality for Caputo fractional derivatives:
	\begin{align*}
	\left(mf^{(n)}\left(\frac{b}{m}\right)+f^{(n)}(a)\right)&\leq \frac{\Gamma(n-\alpha)}{(b-a)^{n-\alpha}}\left((^{C}D^{\beta}_{a^+}f)(b)+(^{C}D^{\alpha}_{b^-}f)(a)\right)\\&\leq\frac{2}{(m+1)(n-\alpha)h\left(\frac{1}{2}\right)}f^{(n)}\left(\frac{a+b}{2}\right).\nonumber
	\end{align*}	
\end{corollary}
\begin{remark}
	Axioms (i)-(vii) of Remark \ref{rem2} for Theorem \ref{thm1} are valid for Theorem \ref{thm3}.
\end{remark}
\indent Next we give the following results as an application of previous established results. First we apply Theorem \ref{thm1}, and get the following result.
\begin{theorem}
	Under the assumptions of Theorem \ref{thm1}, we have
	\begin{align}\label{11*}
	& \left(^{C}D_{a^{+}}^{\alpha-1}f\right)(b)+\left(^{C}D_{b^{-}}^{\beta-1}f\right)(a)\\&\leq \left(\frac{(b-a)^{n-\alpha+1}f^{(n)}(a)}{\Gamma(n-\alpha+1)}+\frac{(b-a)^{n-\beta+1}f^{(n)}(b)}{\Gamma(n-\beta+1)}\right.\nonumber\\&\left.+m\left(f^{(n)}\left(\frac{a}{m}\right)\frac{(b-a)^{n-\beta+1}}{\Gamma(n-\beta+1)}+f^{(n)}\left(\frac{b}{m}\right)\frac{(b-a)^{n-\alpha+1}}{\Gamma(n-\alpha+1)}\right)\right)\int_{0}^{1}h(z)dz.\nonumber
	\end{align}
\end{theorem}
\begin{proof}
	If we take $x=a$ in (\ref{2.1}), then we get following inequality
	\begin{equation}\label{4*}
	\left(^{C}D_{b^{-}}^{\beta-1}f\right)(a)\leq\frac{(b-a)^{n-\beta+1}}{\Gamma(n-\beta+1)}\left(f^{(n)}(b)+mf^{(n)}\left(\frac{a}{m}\right)\right)\int_{0}^{1}h(z)dz.
	\end{equation}
	If we take $x=b$ in (\ref{2.1}), then we get following inequality
	\begin{equation}\label{3*}
	\left(^{C}D_{a^{+}}^{\alpha-1}f\right)(b)\leq\frac{(b-a)^{n-\alpha+1}}{	\Gamma(n-\alpha+1)}\left(f^{(n)}(a)+mf^{(n)}\left(\frac{b}{m}\right)\right)\int_{0}^{1}h(z)dz.
	\end{equation}
	Adding (\ref{4*}) and (\ref{3*}), we get required inequality in (\ref{11*}).
\end{proof}
\begin{corollary}
	By taking $\alpha=\beta$ in (\ref{11*}), then we get the following inequality for Caputo fractional derivatives:
	\begin{align*}
	&\left(^{C}D_{a^{+}}^{\alpha-1}f\right)(b)+\left(^{C}D_{b^{-}}^{\alpha-1}f\right)(a)\\&\leq\frac{(b-a)^{n-\alpha+1}}{\Gamma(n-\alpha+1)} \left(f^{(n)}(a)+f^{(n)}(b)\nonumber+m\left(f^{(n)}\left(\frac{a}{m}\right)+f^{(n)}\left(\frac{b}{m}\right)\right)\right)\\&\times\int_{0}^{1}h(z)dz.\nonumber
	\end{align*}
\end{corollary}
Next we apply Theorem \ref{thm2}, and get the following result.
\begin{theorem}
	Under the assumptions of Theorem \ref{thm2}, we have
	\begin{align}\label{11**}
	&\left|(^{C}D^{\alpha}_{a^{+}}f)(b) + (^{C}D^{\beta}_{b^{-}}f)(a)-\left(\frac{(b-a)^{n-\alpha}f^{(n)}(a)}{\Gamma(n-\alpha + 1)} +\frac{(b-a)^{n-\beta}f^{(n)}(b)}{\Gamma(n-\beta + 1)}\right)\right|\\&\leq \left(\frac{(b-a)^{n-\alpha+1}|f^{(n+1)}(a)|}{\Gamma(n-\alpha + 1)}+\frac{(b-a)^{n-\beta+1}|f^{(n+1)}(b)|}{\Gamma(n-\beta + 1)}\nonumber\right.\\&\left. +m\left(\frac{(b-a)^{n-\alpha+1}}{\Gamma(n-\alpha + 1)}\left|f^{(n+1)}\left(\frac{b}{m}\right)\right|+\frac{(b-a)^{n-\beta+1}}{\Gamma(n-\beta + 1)}\left|f^{(n+1)}\left(\frac{a}{m}\right)\right|\right)\right)\nonumber\\&\times\int^{1}_{0} h(z)dz.\nonumber
	\end{align}
\end{theorem}
\begin{proof}
	If we take $x=a$ in (\ref{0}), then we get following inequality
	\begin{align}\label{4**}
	&\left|(^{C}D^{\beta}_{b^{-}}f)^{(n)}(a)-\frac{f^{(n)}(b)(b-a)^{n-\beta}}{\Gamma(n-\beta+1)}\right|\\&\leq\frac{(b-a)^{n-\beta+1}}{\Gamma(n-\beta+1)}\left(|f^{(n+1)}(b)|+m\left|f^{(n+1)}\left(\frac{a}{m}\right)\right|\right)\int^{1}_{0} h(z)dz.\nonumber
	\end{align}
	If we take $x=b$ in (\ref{0}), then we get following inequality
	\begin{align}\label{3**}
	&\left|(^{C}D^{\alpha}_{a^{+}}f)(b)-\frac{f^{(n)}(a)(b-a)^{n-\alpha}}{\Gamma(n-\alpha+1)}\right|\\&\leq\frac{(b-a)^{n-\alpha+1}}{\Gamma(n-\alpha+1)}\left(|f^{(n+1)}(a)|+m\left|f^{(n+1)}\left(\frac{b}{m}\right)\right|\right)\int^{1}_{0} h(z)dz.\nonumber
	\end{align}
	Adding (\ref{4**}) and (\ref{3**}), we get required inequality in (\ref{11**}).
\end{proof}
\begin{corollary}
	By taking $\alpha=\beta$ in (\ref{11**}), then we get the following inequality for Caputo fractional derivatives:
	\begin{align*}
	&\left|(^{C}D^{\alpha}_{a^{+}}f)(b) + (^{C}D^{\beta}_{b^{-}}f)(a)-\frac{(b-a)^{n-\alpha}}{\Gamma(n-\alpha + 1)}\left(f^{(n)}(a) +f^{(n)}(b)\right)\right|\\&\leq \frac{(b-a)^{n-\alpha+1}}{\Gamma(n-\alpha + 1)}\left(|f^{(n+1)}(a)|+|f^{(n+1)}(b)|\nonumber\right.\\&\left. +m\left(\left|f^{(n+1)}\left(\frac{b}{m}\right)\right|+\left|f^{(n+1)}\left(\frac{a}{m}\right)\right|\right)\right)\int^{1}_{0} h(z)dz.\nonumber
	\end{align*}
\end{corollary}
By applying Theorem \ref{thm3}, similar results can be established, therefore we leave it for reader.
\section*{Concluding remarks}
This paper have been prepared to address Caputo fractional integral inequalities via functions whose nth derivatives are $(h-m)$-convex. It is remarkable to mention that the presented results contain Caputo fractional inequalities for $h$-convex functions, $m$-convex functions, convex functions, Godunova-Levin functions, $p$-functions and $s$-convex functions in second sense on the domain of nonnegative real numbers. Further in application point of the results of this paper may be useful in studying uniqueness of solutions fractional differential equations, analyzing fractional models of different dynamic systems, complex systems etc. In future we will try to study existence of solutions of fractional systems under constraints of such fractional derivative inequalities.  

\vspace{4mm}\noindent{\bf Acknowledgments}\\
\noindent This research work is supported by Higher
Education Commission of Pakistan under NRPU 2016,
Project No. 5421.


% BibTeX users please use one of
%\bibliographystyle{spbasic}      % basic style, author-year citations
%\bibliographystyle{spmpsci}      % mathematics and physical sciences
%\bibliographystyle{spphys}       % APS-like style for physics
%\bibliography{}   % name your BibTeX data base

% Non-BibTeX users please use
\begin{center}

\end{center}



{\small

\noindent{\bf Ghulam Farid}

\noindent Department of Mathematics

\noindent Assistant Professor of Mathematics

\noindent COMSATS University Islamabad, Attock Campus

\noindent Attock, Pakistan

\noindent E-mail: faridphdsms@hotmail.com}\\

{\small
\noindent{\bf  Vishnu Narayan Mishra}

\noindent  Department of Mathematics

\noindent Associate Professor of Mathematics

\noindent Indira Gandhi National Tribal University


\noindent Amarkantak, India

\noindent E-mail: vishnunarayanmishra@gmail.com}\\



\end{document}